Author's response to reviews

Title: Evaluation of the quality of informed consent in a vaccine field trial in a developing country setting

Authors:

Deon Minnies (deon.minnies@uct.ac.za)
Tony Hawkridge (thawkridge@aeras.org)
Willem Hanekom (willem.hanekom@uct.ac.za)
Rodney Ehrlich (rodney.ehrlich@uct.ac.za)
Leslie London (leslie.london@uct.ac.za)
Gregory Hussey (gregory.hussey@uct.ac.za)

Version: 3 Date: 24 September 2008

Author's response to reviews:

Dear Editor,

In this revision I affected the following editorial changes:
1. Corrected the authors list,
2. Removed the keywords and word count section,
3. Incorporated the footnote as a reference into the text,
4. Modified my figures to the format recommended, and corrected the figure order,
5. Moved my tables into the text, and
6. Changed my reference citation to square brackets.

I trust these improvements meet your approval.

Thank you

Deon Minnies